Author’s response to reviews

Title: ADEQUACY OF CONTROL OF CARDIOVASCULAR RISK FACTORS IN AMBULATORY PATIENTS WITH TYPE 2 DIABETES ATTENDING DIABETES OUTPATIENT CLINIC AT A COUNTY HOSPITAL, KENYA

Authors:
Frederick C F Otieno (cfotieno@gmail.com)
Elijah Ogola (onogola@gmail.com)
Mercy Kimando (mkimando@gmail.com)
Kenn Mutai (mutaikk@gmail.com)

Version: 2 Date: 07 Aug 2017

Author’s response to reviews:

The response to reviewers is incorporated in the text and presented in BOLD. There is an accompanying letter attached.